Author’s response to reviews

Title: Peripheral blood lymphocytes immunophenotyping predicts disease activity in clinically isolated syndrome patients

Authors:
Helena Posova (hmare@l1.cuni.cz)
Dana Horakova (dana.horakova@lf1.cuni.cz)
Vaclav Capek (venca@cicconia.cz)
Tomas Uher (tomas.uher@vfn.cz)
Zdenka Hruskova (zdenka.hruskova@vfn.cz)
Eva Havrdova (ehavr@lf1.cuni.cz)

Version: 5 Date: 07 Jun 2017

Author’s response to reviews:

NURL-D-16-00310

Peripheral blood lymphocytes immunophenotyping predicts disease activity in clinically isolated syndrome patients

Helena Posova, M.D., PhD.; Dana Horakova, PhD.; Vaclav Capek; Tomas Uher, PhD; Zdenka Hruskova, PhD; Eva Havrdova

Dear Mr. Chang,

Thank you again for reviewing our manuscript and the opportunity to re-submit a revised version for your consideration. We asked two native speakers (biologist and haematologist) to improve our English.

Yours sincerely,

Helena Posová
Reviewer reports:

MIng-Feng Liao (Reviewer 2): the authors only do one changes of the manuscript. The figures are a little complicate. However, the authors kept the original edition, because authors are afraid of the possibility of losing comprehensiveness of the analysis. I agree with that. However, I suggested the authors can use language editing services before publication.

We did it.